Author's response to reviews

Title: Factors associated with Haemoglobin levels among Bangladeshi pregnant women

Authors:

Hasina Akhter Chowdhury (lima_125@yahoo.com)
Kazi Rumana Ahmed (lopa_birdem@yahoo.com)
Fatema Jebunnessa (runabd2005@yahoo.com)
Jesmin Akter (uttaron_007@yahoo.mail)
Sharmin Hossain (shormybd@yahoo.com)
Md Shahjahan (mdshahjahan@agnionline.com)

Version: 2 Date: 10 April 2014

Author's response to reviews: see over
Editor: Please revise your manuscript to include line and page numbers.

Reply: Line and page numbers are providing in main text.